Author’s response to reviews

Title: Dilemma between health and environmental motives when purchasing animal food products: sociodemographic and nutritional characteristics of consumers

Authors:

Sandrine Péneau (s.peneau@eren.smbh.univ-paris13.fr)
Philippine Fassier (p.fassier@eren.smbh.univ-paris13.fr)
Benjamin Allès (b.alles@eren.smbh.univ-paris13.fr)
Emmanuelle Kesse-Guyot (e.kesse@eren.smbh.univ-paris13.fr)
Serge Hercberg (s.hercberg@eren.smbh.univ-paris13.fr)
Caroline Méjean (c.mejean@eren.smbh.univ-paris13.fr)

Version: 1 Date: 08 Sep 2016

Author’s response to reviews:

Dear editor,

We have made requested modification accordingly. The EudraCT number was already included but we have added the date. We do not have additional TRN number.

Best regards.

Sandrine Péneau